Author's response to reviews

Title: Deprescribing benzodiazepines and Z-drugs in community-dwelling adults: a scoping review

Authors:

André S Pollmann (andre.p@dal.ca)
Andrea L Murphy (andrea.murphy@dal.ca)
Joel C Bergman (joel.bergman@dal.ca)
David M Gardner (david.gardner@dal.ca)

Version: 2  
Date: 30 January 2015

Author's response to reviews:

January 30, 2015

BMC Pharmacology and Toxicology
Editorial Office

Dear BMC Pharmacology and Toxicology editors:

Please consider our transferred submission, “Deprescribing benzodiazepines and Z-drugs in community-dwelling adults: a scoping review”, for publication in BMC Pharmacology and Toxicology.

Reviewers at BMC have suggested that our research may be better suited for this journal. Given the high prevalence of benzodiazepine and Z-drug use and the increasing recognition of the harms of polypharmacy, our submission is timely and relevant to the field of clinical pharmacology.

Thank you for considering our research for review and potential publication.

Yours sincerely,

André Pollmann, on behalf of all authors